Author's response to reviews

Title: Cholera outbreaks (2012) in three districts of Nepal reveal clonal transmission of multi-drug resistant Vibrio cholerae O1

Authors:

Sameer M. Dixit (s.dixit@cmdn.org)
Fatema-Tuz Johura (mbmita@gmail.com)
Sulochana Manandhar (manandhar.sulochana@outlook.com)
Abdus Sadique (sadique@icddrb.org)
Rajesh M. Rajbhandari (r.rajbhandari@cmdn.org)
Shahnewaj B. Mannan (newaj_mb@yahoo.com)
Mahamud-ur Rashid (mahamudur@icddrb.org)
Saiful Islam (saiful_mbd@yahoo.com)
Dibesh Karmacharya (dibesh@cmdn.org)
Haruo Watanabe (haruwata@nih.go.jp)
R. B. Sack (rsack@jhsph.edu)
Alejandro Cravioto (Alejandro.Cravioto@ivi.int)
Munirul Alam (munirul@icddrb.org)

Version: 5
Date: 20 March 2014

Author's response to reviews: see over
To  
Editor,  
BMC Infectious disease  

Subject: Third revision of the manuscript for publication  

Dear Sir/M’m,  
Please find the third revision of the manuscript (MS) entitled “Cholera outbreaks (2012) in three districts of Nepal reveal clonal transmission of multi-drug resistant *Vibrio cholerae* O1” for your kind review and publication. We have added the information that the editor requested. Please see below, our responses to editor is intercalated.  

May I therefore request you kindly to consider our MS for publication in your esteemed journal?  

Sincerely yours,  

Munirul Alam  

Thank you for your email. It has just come to our attention that your ethics and consent statement has accidentally been omitted from the current version of your manuscript. Could you include this information again in your methods section.  

**Our Answer: We have added the ethics and consent statement in the method section.**  

In addition (and apologies that we did not request this earlier) could you deposit your sequence and phylogenetic data in public repositories and include this information in your manuscript.  

**Our Answer: We have deposited the nucleotide sequence data in the GenBank and provided accession numbers for the respective genes in the manuscript.**  

**So far, there is no database to submit the phylogenetic data.**